Patterns of Migration and Risks Associated with Leprosy among Migrants in Maranhão, Brazil

Christine Murto^{1,2}, Frédérique Chammartin^{1,2}, Karolin Schwarz³, Lea Marcia Melo da Costa⁴, Charles Kaplan⁵, Jorg Heukelbach^{6,7}*

1 Swiss Tropical and Public Health Institute, Basel, Switzerland, 2 University of Basel, Basel, Switzerland, 3 University of Cologne, Cologne, Germany, 4 Leprosy Control Program; State Health Secretariat of Maranhão, São Luis, Brazil, 5 University of Southern California, School of Social Work, Hamovitch Center for Science in the Human Services, Los Angeles, California, United States of America, 6 Federal University of Ceará, Fortaleza, Brazil, 7 Anton Breinl Centre for Public Health and Tropical Medicine, James Cook University, Townsville, Australia

Abstract

Leprosy remains a public health problem in Brazil with new case incidence exceeding World Health Organization (WHO) goals in endemic clusters throughout the country. Migration can facilitate movement of disease between endemic and nonendemic areas, and has been considered a possible factor in continued leprosy incidence in Brazil. A study was conducted to investigate migration as a risk factor for leprosy. The study had three aims: (1) examine past five year migration as a risk factor for leprosy, (2) describe and compare geographic and temporal patterns of migration among past 5-year migrants with leprosy and a control group, and (3) examine social determinants of health associated with leprosy among past 5-year migrants. The study implemented a matched case-control design and analysis comparing individuals newly diagnosed with leprosy (n = 340) and a clinically unapparent control group (n = 340) without clinical signs of leprosy, matched for age, sex and location in four endemic municipalities in the state of Maranhão, northeastern Brazil. Fishers exact test was used to conduct bivariate analyses. A multivariate logistic regression analysis was employed to control for possible confounding variables. Eighty cases (23.5%) migrated 5-years prior to diagnosis, and 55 controls (16.2%) migrated 5-years prior to the corresponding case diagnosis. Past 5 year migration was found to be associated with leprosy (OR: 1.59; 95% CI 1.07-2.38; p = 0.02), and remained significantly associated with leprosy after controlling for leprosy contact in the family, household, and family/household contact. Poverty, as well as leprosy contact in the family, household and other leprosy contact, was associated with leprosy among past 5-year migrants in the bivariate analysis. Alcohol consumption was also associated with leprosy, a relevant risk factor in susceptibility to infection that should be explored in future research. Our findings provide insight into patterns of migration to localize focused control efforts in endemic areas with high population mobility.

Citation: Murto C, Chammartin F, Schwarz K, da Costa LMM, Kaplan C, et al. (2013) Patterns of Migration and Risks Associated with Leprosy among Migrants in Maranhão, Brazil. PLoS Negl Trop Dis 7(9): e2422. doi:10.1371/journal.pntd.0002422

Editor: Richard O. Phillips, Kwame Nkrumah University of Science and Technology (KNUST) School of Medical Sciences, Ghana

Received February 17, 2013; Accepted July 29, 2013; Published September 5, 2013

Copyright: © 2013 Murto et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This research was co-financed by the Brazilian Research Council (CNPq, http://www.cnpq.br/) and the Department of Science and Technology of the Brazilian Ministry of Health (DECIT, http://portal.saude.gov.br/portal/saude/profissional/visualizar_texto.cfm?idtxt = 29045&janela = 1). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: heukelbach@web.de

Introduction

Leprosy continues to be an endemic disease in many parts of the world. Brazil has globally the second highest new case incidence [1]. National leprosy prevalence of 1.54/10,000 in 2010 [2] remains above the WHO goal of <1 per 10,000. Highly endemic areas of the disease continue to persist despite large-scale national efforts to control the disease. A challenge in disease control efforts is compounded as leprosy can be diagnosed many years after infection took place due to the long incubation period, and mild early symptoms of the disease may be overlooked. Migration has been found to be a social determinant of disease [3], and has been hypothesized as a risk factor in continued leprosy incidence [4,5,6]. In fact, earlier research in Brazil highlighted the increased distribution of leprosy along new corridors coinciding with frontier expansion connecting southern agricultural areas to the north of Brazil [7], as well as periurban migrant settlements on the outskirts of urban centers [4]. Migrants move between endemic and nonendemic areas in Brazil and often live in substandard conditions.

As an infectious disease caused by *Mycobacterium leprae*, leprosy primarily affects the skin and peripheral nerves and causing sensory loss. While nasal mucosa is considered the main transmission site, new research indicates that oral presence of *M.leprae* bacilli may be an additional mode of transmission [8]. Maranhão, the study area of this research, has the third highest prevalence of leprosy (5.34/10,000) in the country [2] and is among the states with the highest out- and return- migration rates [9].

The proliferation of leprosy in Brazil continues largely in conditions of poverty that include poor housing and sanitation, high household density, illiteracy and low socioeconomic levels both at the micro and macro levels [4,10–13]. Rapid population growth and uncontrolled urbanization, often as a consequence of migration for employment and differential access to services between rural and urban areas, has facilitated the expansion of these poor social and environmental conditions on the peripheries of cities associated with leprosy infection [4–5,7,13]. Additionally, new road construction and railways have enabled movement

Author Summary

In Brazil, leprosy remains a significant public health problem in endemic clusters of high transmission risk throughout the country. Migration is thought to be a factor associated with continued leprosy transmission, as migration has also been found to be associated with other Neglected Tropical Diseases (NTDs). We analyzed the association between past five year migration and leprosy as part of a larger epidemiological study evaluating risk factors for infection among recently diagnosed leprosy cases (n = 340) and a matched clinically unapparent control group (n=340) in the northeastern state of Maranhão. Among migrants with leprosy, 23.5% (n = 80) migrated in the past five years, with 16.2% (n = 55) of the control group. Past five year migration was significantly associated with leprosy, and remained significant after controlling for household and familial contact as potential confounders. Factors found to be associated with leprosy among past 5year migrants included alcohol consumption, poverty, and household, family and other leprosy contact. Key patterns of movement emerged from the study that may aid future regional leprosy control efforts.

between rural communities and urban areas. These developments in transportation have been argued to explain the expanded distribution of leprosy in Brazil [4–6]. Nevertheless, household leprosy contact continues to be the primary risk factor associated with leprosy infection [14]. Proximity to the household contact has been seen as relevant in terms of increased risk [15]. Consanguineous contact has also been found to be associated with leprosy. Findings from Moet et al. (2006) suggest evidence of a genetic relationship independent of physical contact for leprosy infection.

Migration has been found to be an impediment to both leprosy elimination and control efforts. Prior research has suggested that migration may influence transmission and distribution of the disease [5,16] as well as other neglected tropical diseases (NTDs) [3,17–23]. This study explores the spatial and temporal patterns of migration in individuals with leprosy in Maranhão. The study also examines risk factors associated with leprosy among individuals who have migrated in the past five years (past 5-year migrants). Comparison of risks associated with leprosy and migration is challenging in a homogeneous population. However evaluation of specific risk factors that differentiate leprosy among past 5-year migrants from a clinically unapparent control group without clinical signs of leprosy who migrated in the past five years in this investigation, sheds light on those factors that are of importance when considering leprosy infection and expression of disease. The study has three specific aims: 1) to examine if migration in the past five years is a risk factor for leprosy; 2) to describe and compare geographic and temporal patterns of migration among past 5-year migrants with leprosy and a control group without clinical signs of leprosy; 3) to examine the social determinants of health associated with leprosy among past 5-year migrants.

Methods

Ethics statement

Written approval was obtained from the Ethical Review Board of the Federal University of Ceará (Fortaleza, Brazil). Permission to perform the study was also obtained by the Maranhão State Health Secretariat, the State Leprosy Control Program and municipalities involved. Informed written consent was obtained from study participants, or their parent/guardian in the case of minors, after explaining the objectives of the study. Interviews were conducted in private.

Study area

The research was conducted in four leprosy endemic municipalities in the state of Maranhão, Brazil: Santa Inês, São José de Ribamar, Codó, and Bacabal. These municipalities are located in a major endemic cluster identified by the Brazilian Ministry of Health as a high-risk area for leprosy transmission [16]. Santa Inês, (population 77,282) [24], Codó (population 118,038) [24], and Bacabal (population 100,014 same) [24] are small townships in the interior of Maranhão that are largely surrounded by rural agricultural production, while São José de Ribamar (population 163,045) [24] is on the outskirts of the capital city, São Luis. Most households are small brick or mud and palm residences with rudimentary plumbing and hammocks to accommodate the multigenerational inhabitants.

Study design

A case-control study was designed as part of an extended epidemiological investigation on risk factors associated with leprosy infection in four highly endemic municipalities in Maranhão, as part of the MAPATOPI study. The MAPATOPI study is an interdisciplinary project to support and improve the Brazilian leprosy program in Maranhão, Pará, Tocantins, and Piaui. Variables associated with past five year migration among those diagnosed with leprosy between 2009-2010 were compared with a matched clinically unapparent control group without clinical signs of leprosy. Migration was defined as those who resided outside of the municipality of their current residence, and is limited to five years as this is the average incubation period from leprosy infection to symptom onset. Past five year migration data is also collected in the Brazilian National Household Survey [9]. A detailed analysis of socio-cultural, health service related and economic variables that were collected as part of the larger epidemiological study will be explored elsewhere.

Study sample

The case group was identified through the database of the National Information System for Notifiable Diseases (Sistema de Informação de Agravos de Notificação - SINAN) and included adults 15 and older in each of the four sites diagnosed with leprosy in 2009-2010 (n = 394). Individuals under 15 years of age, those previously diagnosed with leprosy and relapsed, living outside of the highly endemic cluster and who could not be located through multiple contact attempts were excluded from the study. The control group (n = 391) was selected from the Programa Saúde da Família (Program for Family Health). This program registers all families in the catchment areas of the clinic by community health workers. At the clinics, we randomly selected intake forms from the Program for Family Health for age and sex at each clinic and contacted those individuals for inclusion in the control group. Each of the matched controls were clinically evaluated for signs of leprosy. Any individual with a clinical suspicion of leprosy was excluded from the study and referred to municipal health centers for further diagnostic testing.

Data collection

Data collection was conducted between April and August 2010. Data collection was coordinated through the Municipality Health Secretariats with the support of the Maranhão State Health Secretariat. Study participants were recruited by community health agents for the study. They were interviewed by trained health professionals at the local health care centers, or in patient homes when disability or age prevented health center attendance. Information on demographics, socioeconomic status, healthcare access, migration, behavior and stress was collected through structured questionnaires. Clinical data were also collected through patient medical records.

Data analysis

Data were entered twice using EpiInfo software version 3.5.1 (Centers for Disease Control and Prevention, Atlanta, USA) and cross-checked for entry-related errors. Statistical tests were used to assess normality. Included are data sets with information related to migration. Any cases that did not have complete migration data were excluded from the analysis. Of the 340 leprosy cases and 340 matched controls, we first identified 135 (19.9%) past 5-year migrants in the case (n = 80) and control groups (n = 55). The distribution of key demographic, spatial and temporal migration pattern variables among past 5-year migrants in the case and control groups was examined and tested by the use of Fishers exact test for significant differences in the stratified sample of past 5-year migrants.

We then conducted bivariate analyses comparing cases (n = 340) and controls (n = 340) using Fishers exact test to examine if past five year migration was associated with leprosy diagnosis. As household contact remains the most significant known transmission risk to date for leprosy infection [14,15], we additionally undertook multivariate logistic regression analysis controlling for family (parent, child and/or sibling) and household (consanguineous and/or non-consanguineous) contact with leprosy.

Next, stratified bivariate analyses using Fishers exact tests were used to determine differences in the association among social determinants of health (socioeconomic status), psychosocial (alcohol use and life stressors) and biosocial factors (leprosy contact exposure) for case and control groups of past 5-year migrants (n = 135).

Results

A total of 394 leprosy cases and 391 controls were interviewed. There were 23 relapsed leprosy cases and 12 controls suspected of leprosy who were excluded from the study. Eight respondents refused to participate. Complete migration data was available for 680 respondents. Of the 340 leprosy cases and 340 matched clinically unapparent controls, 23.5% of those with leprosy (n = 80 cases) and 16.2% (n = 55) of the control group without clinical signs of leprosy migrated in the past 5 years before diagnosis. Only 4.4% (n = 15) of cases migrated after diagnosis. Table 1 reflects migration into and out of major endemic clusters identified by the Brazilian Ministry of Health as high-risk areas for leprosy transmission [6] (Figure 1), and other demographics and migration variables. These variables were not significantly associated with leprosy among past 5-year migrants prior to diagnosis (test results not shown). Leprosy cases were largely among the youngest age group (15–29) migrating, with an equal distribution between males and females. More than one-third of those with leprosy who migrated in the past five years were illiterate. The majority of leprosy cases migrated within cluster 1, which includes the northern states of Pará, Piauí, Tocantins and Maranhão. More than half (56.3%) of cases moved between municipalities in Maranhão, followed with fewer cases to neighboring Pará (11.8%), Piauí (3.9%) and Tocantins (2.0%), and onefifth of migrants were drawn to non-contiguous states. All those with leprosy migrated into a highly endemic cluster on at least one occasion, not including their current residence.

Nearly one in six migrants with leprosy migrated for employment in the last five years and this was slightly less than expected for internal population movement. Typical of internal population flow, most migration in Maranhão was to urban areas (60.3%) compared to rural areas (33.3%), and both rural and urban areas (7.7%). Social networks in migration destination sites for those with leprosy had a higher tendency to be family contacts with whom they lived (81.0%) than work contacts (17.7%). This may be an explanation for the significant number of respondents who always had a contact prior to migrating (79.8%). Migrants with leprosy lived on average with 8.61 people per household while migrating.

Past five year migration prior to diagnosis was found to be significantly associated with leprosy as shown in Table 2 which represents the results of the multivariate logistic regression analysis. Past five year migration remained significantly associated with leprosy after controlling in separate models for 1) household contact (consanguineous and/or non-consanguineous); 2) family contact (parent, child and/or sibling; 3) and household and family contact in multiple logistic regression models.

Key social, biosocial, and behavioral factors were found to be associated with leprosy (Table 3). Household, familial and other contact with someone infected with leprosy was significantly different for leprosy infected past 5-year migrants compared to control group migrants. Genetic association of closely related kinship shows a significant difference for contact with parent/child/sibling (OR: 7.82; CI 95%: 2.32-33.38; P-value = 0.0001). Contact regardless of consanguinity (OR: 4.99; CI 95%: 1.7-16.51; P-value = 0.001) and actual household contact (OR: 5.54; CI 95%: 1.49-30.46; P-value = 0.004) was also significant. An important behavioral factor distinguishing migrants with leprosy compared to the clinically unapparent control group was past five year alcohol consumption (OR: 4.46; CI 95%: 1.43-14.15; P-value = 0.005).

Income and other socioeconomic variables showed significant differences between migrants with leprosy and the control group. Income less than the minimum wage (OR: 2.12; CI 95%: 0.97–4.71; p-value = 0.049) as well as poor access to public waste services (OR: 3.1; CI 95%: 1.1–10.02; p-value = 0.03) and family illiteracy (OR: 2.67; CI 95%: 1.13–6.51; p-value = 0.02) were found to be associated with leprosy among past 5-year migrants.

Education, presence of BCG scar, zone of residence and lifestyle stressors - separation from family and friends, loss of employment or income, marital separation or death of close friend or relativewere not significantly associated with leprosy among past five year migrants.

Discussion

Leprosy was introduced to Brazil through European colonization and later through slave movement so that by the 1600's, leprosy was well established in the country [25]. More recently, migration has been hypothesized to be an impediment to leprosy control, and spatial analysis indicates the introduction of leprosy through inter and intra-state population movement in Brazil [5], as well as expanded distribution of leprosy through migration [26]. Population movement can put both migrants and non-migrants at risk when diseases move between endemic and non-endemic areas. Latent symptomology, characteristic of leprosy, could facilitate the distribution of disease when no symptoms are present, or when mild symptoms are overlooked. The migrant lifestyle poses similar marginalized socioeconomic, behavioral and environmental risks that have been well established as factors associated with leprosy transmission [4,10–13,27–28].

Leprosy in this study was found to be significantly associated with past five year migration. Susceptibility among migrants may, in part, be due to spatial and temporal patterns of movement in Table 1. Demographics and migration patterns of past 5-year migrant leprosy cases and clinically unapparent controls.

	Leprosy Cases		Controls	
	Included*(n = 80)†	%	Included (n = 55)†	%
Demographics				
Age				
15–29	35	43.8	28	50.9
30–44	21	26.3	14	25.5
45–59	15	18.8	9	16.4
60 or older	9	11.3	4	7.3
Gender				
Male	40	50.0	35	63.6
Female	40	50.0	20	36.4
Education				
Literate	54	67.5	45	81.8
Illiterate	26	32.5	10	18.2
Migration Patterns				
Leprosy Cluster Migration				
Cluster 1	48	60.0	32	58.2
Cluster 2	3	3.8	1	1.8
Cluster 6	1	1.3	0	0
Cluster 7	1	1.3	1	1.8
Cluster 9	2	2.5	0	0
Out of cluster migration	25	31.3	21	38.2
Migration in cluster				
1 time	71	88.8	49	89.1
2 or more times	9	11.3	5	9.1
In-state vs. out of state migration				
In Maranhão	45	56.3	25	45.5
Other state	35	43.75	30	54.6
No. of times migrated past 5-yrs				
1 time	61	76.3	47	85.5
2 or more times	19	23.8	8	14.5
Zone of migration in past 5-yrs				
Urban only	47	60.3	38	70.4
Rural only	26	33.3	13	24.1
Rural and Urban	6	7.7	3	5.6
Migration for work in past 5-yrs				
Yes	46	57.5	30	55.6
No	34	42.5	25	45.5
Social network prior to migration				
Always	63	79.8	39	70.9
Sometimes	5	6.3	1	1.8
Never	11	13.9	15	27.3
Who lived with during migration				
Family	64	81.0	41	74.6
Co-workers	14	17.7	12	21.8
Other	1	1.3	2	3.6
Mean $\#$ people lived with during migration	80	8.61	55	6.7
Mean years of migration		6.25		4.8

[†]Data not available for all individuals. doi:10.1371/journal.pntd.0002422.t001



Figure 1. Locations of the 10 most probable leprosy clusters (yellow regions) and municipal councils (dots), Brazil, 2005–2007. [6]. doi:10.1371/journal.pntd.0002422.g001

and between areas identified by the Brazilian Ministry of Health as highly endemic clusters for leprosy transmission [16]. While we found no significant difference between key spatial and temporal variables and past five year migration among those with leprosy compared to the clinically unapparent control group, more than half of movement for internal migration among those with leprosy was within the leprosy endemic cluster in the state of Maranhão. Few migrated to the other nine endemic clusters in Brazil, a third migrated to other non-endemic areas, and less than half migrated outside of Maranhão. From an operational perspective for leprosy control in Brazil, this provides sufficient evidence to suggest future surveillance of population flow between municipalities in Maranhão, which should involve comparison of the distribution of leprosy incidence over the five year latency period. Should these areas be identified as emerging endemic areas, service delivery strategies should target these as focal points for state control efforts.

Maranhão continues to be a state with higher net out- and return- migration [9]. Interstate population movement, such as to neighboring Pará, draws many poor migrants from Maranhão's interior leprosy endemic areas to the employment found in largescale mining and agriculture industries [29–30]. Interstate movement necessitates cross-border cooperation for leprosy control and may aid in identifying impending high-risk areas for disease distribution. In fact, other research showed that 5.2% of leprosy patients in *Cluster 1* (including Maranhão and neighboring states of Tocantins, Piaui and Pará) were diagnosed outside of their municipality of residence between 2001–2009 [31]. Municipalities in Maranhão and neighboring Pará, which have the third and fifth highest new case incidence in the country respectively, would be good targets for future collaborative surveillance projects.

Our findings indicate that the majority of migration in Maranhão continues to be between rural and urban areas, consistent with other research on population flow in Brazil [32]. However population movement documented in our study appears to be of longer duration than is typical for temporary circular migration. Rural to urban migration is a common solution to reduce poverty, as more and regular job opportunities tend to exist in urban areas [33–35]. This often places migrants at higher risk for disease morbidity and mortality due to poor living conditions in urban slums [36]. Kerr-Pontes et al.'s (2004) [4] ecological study in Brazil's northeast demonstrated that urban population growth due to uncontrolled urbanization and migrant influx from Brazil's rural interior, was a predictor of leprosy incidence.

We found that population movement is clearly facilitated through strong destination based social networks as a precursor to migration. These social networks tend to be family-based, **Table 2.** Crude (OR) and adjusted odds ratios (AOR) for the association of leprosy and five year migration prior to leprosy diagnosis, controlling for household, family, and household and family leprosy contact.

					AOR Controlling for leprosy contact			
	Included (n = 680)	Leprosy Cases N (%)	Controls N (%)	OR (95% CI)	Household contact	Family contact	Household/Family contact	
Past five year migration								
Yes	135	80 (59.3)	55 (40.7)	1.59 (1.07–2.38)*	1.54 (1.03–2.29)*	1.51 (1.01–2.27)*	1.51 (1.0–2.28)**	
No	545	260 (47.7)	285 (52.3)	1.0	1.0	1.0	1.0	

*P<.05. **P<.10.

doi:10.1371/journal.pntd.0002422.t002

as indicated by migrant co-habitation arrangements. On an individual level, social networks enable population movement by reducing the cost of migration through benefits such as established shared housing and employment networks, thus making migration a more attractive option to pursue. On a community level, social networks that facilitate migration can have a cumulative effect in sending municipalities to perpetuate and build upon migrant flow between origin and destination sites [37]. Because of the social nature of these community relationships to kinship, friendships and working relationships, migration can be highly localized to movement between specific neighborhoods in sending and receiving communities. Short-term movement, as Skeldon (2003) [38] points out, is less likely to be measured through census surveys, thus monitoring population movement should be undertaken at the municipality level and integrated into larger databases to establish early warning systems.

Exposure to an index patient has been identified as the primary determinant of leprosy infection among their contacts. The magnitude of the effect of contact in our study was highest among close family contact – parent, child, and/or siblings - followed by consanguineous and/or non-consanguineous household contact and lastly other contact, which could include social and distant family exposure. The possibility of genetic susceptibility to leprosy

Table 3. Factors associated with leprosy diagnosis among past five year migrant cases and clinically unapparent controls.

Social and Behavioral Variables	Included (n = 135) $\dagger \dagger$	Leprosy Cases N (%)	Controls N (%)	OR (95% CI)	P-value
Alcohol consumption					
Never drank	29	15 (51.72)	14 (48.28)	Reference	
Drink currently	43	15 (34.88)	28 (65.12)	0.5 (0.17–1.45)	0.22
Drank in past 5 years	52	43 (82.69)	9 (17.31)	4.46 (1.43–14.15)	0.005
Stopped drinking more than 5 years ago	11	7 (63.64)	4 (36.36)	1.63 (0.32–9.25)	0.72
Leprosy Contact					
Familial and non-familial contact					
No leprosy contact	76	33 (43.42)	43 (56.58)	Reference	
Parent/Child/Sibling with leprosy	28	24 (85.71)	4 (14.29)	7.82 (2.32–33.38)	0.0001
Others with leprosy	29	23 (79.31)	6 (20.69)	4.99 (1.7–16.51)	0.001
Household contact with leprosy past 5/6 years					
Yes	23	20 (86.96)	3 (13.04)	5.54 (1.49–30.46)	0.004
No	108	59 (54.63)	49 (45.37)	Reference	
Socio-economic factors					
Income†					
<=R\$510	55	38 (69.09)	17 (30.91)	2.12 (0.97–4.71)	0.049
>R\$510	76	39 (51.32)	37 (48.68)	Reference	
Public Waste Service					
Yes	107	58 (54.21)	49 (45.79)	Reference	0.03
No	28	22 (78.57)	6 (21.43)	3.1 (1.1–10.02)	
Family Illiteracy					
Yes	44	32 (72.73)	12 (27.27)	2.67 (1.13–6.51)	
No	78	39 (50.0)	39 (50.0)	Reference	0.02

[†]At the time of the survey 1US\$ was equivalent to 1.72R\$, and R\$ 511,- the official minimum wage as set by the Federal Government.

^{††}Data not available for all individuals.

doi:10.1371/journal.pntd.0002422.t003

infection, through close family kinship has been significantly associated with leprosy among contacts [14-15,39] which supports our findings of leprosy association with close kinship among past 5year migrants. At the household level, other research has shown that proximity to and intensity of exposure to leprosy increases the risk of transmission, as much as five to nine times that of nonhousehold contacts [14-15,39-42], although leprosy clustering among neighboring residences in areas of high population density and poverty has social contact risk similar to household contacts [43]. Contact with multibacillary diagnosis in the household has also been associated with increased risk [14-15,41-42] and indicates late diagnosis and long-term exposure to contacts. As the majority of migrants in our sample were diagnosed with multibacillary leprosy, this has significant implications for transmission and also for leprosy associated complications and disability.

Migration was significantly associated with leprosy in our logistic regression models controlling for household and close family contact independently. The independent association with household and close consanguineous exposure could indicate some relationship to familial social networks in migrant destination sites. This, in addition to intensity of exposure due to high household density during migration, suggests both the genetic relationships and social environment surrounding migration may figure prominently in explaining leprosy diagnosis.

The majority of individuals in contact with an index patient are not susceptible to the disease. As such, Sales et al. (2011) [14] suggest that leprosy surveillance should explore multiple factors that may contribute to the risk for infection. While many behavioral, demographic, and socio-environmental variables were included in the analysis, we found socioeconomic status and past five year alcohol consumption among migrants with leprosy were significantly associated with leprosy in comparison to clinically unapparent migrants in the control group. Brazil has one of the highest alcohol-attributable disability-adjusted life years (DALYs) in the world. According to the World Health Organization, there is evidence for an association between alcohol consumption and infectious disease [44]. Current alcohol use however was not significant. This may be the result of recently diagnosed migrants abstaining from alcohol use due to multi-drug therapy treatment. A substantial concern, however was that nearly one in five migrants with leprosy were currently drinking alcohol, which has been associated with leprosy relapse in Brazil [12]. Alcohol consumption can interact with medication absorption [45] and could render leprosy treatment less effective. This can contribute to the elevation of risk for transmission to exposed contacts.

Low socioeconomic status was additionally associated with leprosy among past 5-year migrants. Other research in Cluster 1 also found poverty associated with migration prior to diagnosis among those with leprosy (unpublished data). While poverty is ubiquitously associated with leprosy throughout the literature, it should be noted that these results were taken after the migration period and thus may not be an adequate measure of socioeconomic level during migration. Low socioeconomic status among migrants with leprosy may be linked to restricted employment as the result of disability due to leprosy, or difficulty in sustaining employment during treatment. Despite this, family illiteracy and inaccessibility to public waste collection, proxies for low socioeconomic status in Brazil, were significantly higher for migrants with leprosy compared to the control group. Socioeconomic status, the primary social determinant of health, should be the topic of further investigation both during and after migration.

Conclusion

Leprosy was found to be associated with past five year migration, even after controlling for confounders. In the comparison of past 5-year migrants, leprosy was associated with both household consanguineous and/or non-consanguineous contact, close family and other social leprosy contact, consistent with research identifying contact exposure as the major determinant of leprosy transmission [14–15]. However, the magnitude of effect for leprosy among migrants in our study was most significant among close family and household contacts. As migration in Maranhão was largely facilitated through family networks, contact surveillance should include migration site residence contacts as well as current residence contacts.

While patterns of migration, including movement in and between highly endemic clusters, were not different among migrants with leprosy and clinically unapparent migrants in the control group, important facets of migration emerged that could benefit leprosy control at the state and national level. State control programs should consider monitoring past five year residence among those newly diagnosed with leprosy to identify intra- and inter-state migration flow. This may provide early warning systems for localized disease control in areas yet to be identified as highrisk areas.

Alcohol consumption in the years prior to diagnosis may be associated with susceptibility to leprosy. Alcohol consumption and consumption frequency should be included in future investigations. This research will help to determine the extent that alcohol consumption plays a role in the dynamics of both transmission and expression of leprosy. As alcohol consumption has also been associated with leprosy relapse, further attention should be given to alcohol consumption during treatment, patient relapse and contact exposure to leprosy. Other substances should also be given attention in future research.

Other research in Brazil has found a spatial relationship to migration and distribution of leprosy and an association of leprosy with poor socio-economic conditions [4–6]. Our research shows that in endemic areas leprosy is not only associated with population movement itself, but, most importantly with the social conditions of the migrant in the endemic areas, their behavior, and contact with leprosy in the family and household.

Supporting Information

Checklist S1 STROBE checklist. (DOC)

Acknowledgments

We thank the entire team of the Leprosy Control Program and the State Health Secretariat of Maranhão. Collaboration of the Municipalities' Health Secretariats and Primary Health Care Centers is acknowledged. Special thanks to Professor Marcel Tanner, Director of the Swiss Tropical and Public Health Institute (SwissTPH) for guidance in the methodological approach and final review, Dr. Penelope Vounatsou (SwissTPH) for guidance on the statistical analysis, and Dirk Keidel of SwissTPH for assistance with data management. We are most grateful to all patients that kindly agreed to participate in the study.

This publication is part of the MAPATOPI study, an interdisciplinary project providing evidence for improving the Brazilian leprosy control program.

Author Contributions

Conceived and designed the experiments: CM KS LMMdC JH. Performed the experiments: CM FC KS JH. Analyzed the data: CM FC CK JH. Contributed reagents/materials/analysis tools: CM FC KS CK JH LMMdC. Wrote the paper: CM CK JH LMMdC FC KS.

References

- WHO (2011) Leprosy Update 2011. Weekly Epidemiological Record 36: 389– 400.
- Brasil MOH (2010) Indicadores de morbidade e factores de risco. Taxa de prevalencia de hanseniase. http://tabnet.datasus.gov.br/cgi/deftohtm. exe?idb2011/d09.def.
- Aagaard-Hansen J, Nombela N, Alvar J (2010) Population movement: a key factor in the epidemiology of neglected tropical diseases. Trop Med Int Health 15(11): 1281–1288.
- Kerr-Pontes L, Montenegro A, Barreto M, Werneck G, Feldmeier H (2004) Inequality and leprosy in Northeast Brazil: an ecological study. Int J Epidemiol 33: 262–69.
- Magalhães MCC, Rojas LI (2007) Spatial Differentiation of Leprosy in Brazil. Epidemiologia e Serviços de Saúde 2007 16(2): 75–84.
- Penna ML, Oliveira M, Penna O (2009) Spatial distribution of leprosy in the Amazon region of Brazil. Emer Inf Dis 15(4): 650–52.
- Martelli CM, Moraes Neto OL, Andrade AL, Silva SA, Silva IM, et al (1995) Spatial patterns of leprosy in an urban area of central Brazil. Bull World Health Organ 73(3):315–9.
- Martinez TS, Figueira MM, Costa AV, Gonçalves MA, Goulart LR, et al. (2011) Oral mucosa as a source of Mycobacterium leprae infection and transmission, and implications of bacterial DNA detection and the immunological status. Clin Microbiol Infect 11:1653–8.
- Instituto Brasileiro de Geografia e Estatistica (IBGE) (2009) Pesquisa Nacional por Amostra de Domicílios (PNAD). http://www.ibge.gov.br
- Cury MR, Paschoal VD, Nardi SM, Chierotti AP, Rodrigues Júnior AL, et al (2012) Spatial analysis of leprosy incidence and associated socioeconomic factors. Rev Saúde Pública 46(1): 110–118.
- Kerr-Pontes L, Barreto ML, Evangelista CMN, Rodrigues LC, Heukelbach J, et al.(2006) Socioeconomic, environmental and behavioral risk factors for leprosy in North-east Brazil: results of a case-control study. Int J Epidemiol 35(4): 994– 1000.
- Ferreira SM, Ignotti E, Gamba MA (2011) Factors associated to relapse of leprosy in Mato Grosso, Central-Western Brazil. Rev Saude Publica 45(4): 756– 64.
- Montenegro A, Werneck G, Kerr-Pontes L, Feldmeier H (2004) Spatial analysis of the distribution of leprosy in the state of Ceara, Northeast Brazil. Mem Inst Oswaldo Cruz 99(7): 683–86.
- Sales AM, Leon AP, Duppre NC, Hacker MA, Nery JA, et al. (2011) Leprosy among patient contacts: A multilevel study of risk factors. PLOS NTD 5(3): e1013.
- Moet F, Pahan D, Schuring R, Oskam L, Richardus J (2006) Physical distance, genetic relationship, age and leprosy classification are independent risk factors for leprosy in contacts of patients with leprosy. J of Infec Dis 193: 346–53.
- Penna ML, Penna GO (2009) The epidemiological behavior of leprosy in Brazil. Lep Rev 80(3): 332–44.
- Watts S (2006) The Social Determinants of Schistosomiasis. In: UNICEF/ UNDP/World Bank/WHO Special Programme for Research and Training in Tropical Diseases. Geneva: World Health Organization. 4–90.
- Kloos H, Correa-Oliveira R, Reis DC, Rodriques EW, Monteiro LAS et al. (2010) The role of population movement in the epidemiology and control of schistosomiasis in Brazil: a preliminary typology of population movement. Mem Inst Oswaldo Cruz 105(4): 578–86.
- Drumond JA, Marcopito LF (2006) Internal migration and distribution of chagas disease mortality, Brazil 1981–1998. Cad Saude Publica 22: 2131–40.
- Bedoya-Pacheco SJ, Araujo-Melo MH, Valete-Rosalino CM, Pimentel MI, Conceição-Silva F, et al. (2011) Endemic tegumentary leishmaniasis in Brazil: correlation between level of endemicity and number of cases of mucosal disease. Am J Trop Med Hyg 84(6): 901–5.
- Coura JR, Junqueira AC (2012) Risks of endemicity, morbidity and perspectives regarding the control of Chagas disease in the Amazon Region. Mem Inst Oswaldo Cruz 107(2): 145–54.

- Drumond KO, Costa FA (2011) Forty years of visceral leishmaniasis in the State of Piaui: a review. Rev Inst Med Trop Sao Paulo 53(1): 3–11.
- da Silva-Nunes M, de Souza VA, Pannuti CS, Sperança MA, Terzian Al, et al. (2008) Risk factors for dengue virus infection in rural Amazonia: populationbased cross-sectional surveys. Am J Trop Med Hyg 79(4): 485–94.
- Instituto Brasileiro de Geografia e Estatistica (IBGE) (2010) IBGE Cities. (Accessed 2012) Available at: http://www.ibge.gov.br/cidadesat/topwindow. htm?1
- Scott, Sir HH (1943) The influence of the slave trade on the spread of tropical disease.T Roy Soc Trop Med H 37(3): 169–188.
- Opromolla PA, Dalben I, Cardim M (2006) Geostatistical analysis of leprosy cases in the state of São Paulo, 1991–2002. Rev Saude Publica 40(5):907–13.
- Lustosa AA, Nogueira LT, Pedrosa JI, Teles JB, Campelo V (2011) The impact of leprosy on health-related quality of life. Rev Soc Bras Med Trop 44(5): 621–6.
- Silva DRX, Ignotti E, Souza-Santos R, Hacon SS (2010) Hanseníase, condições sociais e desmatamento na Amazonia brasileira. Rev Panam Salud Publica 27(4): 268–75.
- 29. Barros CJ (2007) Trem de maranhense. Reporter Brasil.
- Fearnside P (2008) The roles and movements of actors in the deforestation of Brazilian Amazonia. Ecology and Society 13(1): 23.
- Alencar CH, Ramos AN Jr, Sena Neto SA, Murto C, Alencar Mde J, et al. (2012) Diagnosis of leprosy in municipalities other than the patients' residence: spatial analysis, 2001–2009. Cad Saude Publica 28(9):1685–98.
- Golgher A, Rosa CH, Araujo Jr AF (2008) Determinants of migration in Brazil: Regional Polarization and Povety Traps. Papeles de Poblacion(56): 135–171.
- 33. Deshingkar P (2008) Circular Internal Migration and Development in India. In: DeWind J, Holdaway J. editors. Migration and Development Within and Across Borders: Reserach and Policy Perspectives on Internal and International Migration. Geneva: IOM and SSRC: 161–187.
- Hossain MZ (2001) Rural-Urban Migration in Bangladesh: A Micro-Level Study. In: IUSSP (Aug 20–24).
- IOM (2005) Internal Migration and Development: A Global Perspective. Geneva: International Organization for Migration (IOM).
- Islam MM, Azad KM (2008) Rural-urban migration and child survival in urban Bangladesh: Are the urban migrants and poor disadvantaged? J Biosoc Sci 40: 83–96.
- Massey DS (1990) Social Structure, Household Strategies, and the Cumulative Causation of Migration. Population Index 56: 3–26.
- Skeldon R (2003) Migration and Poverty. In African Migration and Urbanization in Comparative Perspective. Johannesburg, South Africa. June 4–7.
- Durães SM, Guedes LS, Cunha MD, Magnanini MM, Oliveira ML (2010) Epidemiologic study of 107 cases of families with leprosy in Duque deCaxias, Rio de Janeiro, Brazil. An Bras Dermatol 85(3): 339–45.
- Fischer E, De Vlas S, Meima A, Habbema D, Richardus J (2010) Different Mechanisms for Heterogeneity in Leprosy Susceptibility Can Explain Disease Clustering within Households. PLoS One 5(11): e14061.
- Fine P, Sterne J, Ponninghaus J, Bliss L, Saul J, et al. (1997) Household and dwelling contact as risk factors for leprosy in Northern Malawi. Am J Epidemiol 146(1): 91–102.
- van Beers S, Hatta M, Klatser P (1999) Patient contact is the major determinant in incident leprosy: implications for future control. Int J Lepr Other Mycobact Dis 67(2): 119–28.
- Moura MLN, Dupnik KM, Sampaio GAA, Nóbrega PFC, Jeronimo AK, et al. (2013) Active Surveillance of Hansen's Disease (Leprosy): Importance for Case Finding among Extra-domiciliary Contacts. PLoS Negl Trop Dis 7(3): e2093.
- WHO (2011) Global status report on alcohol and health. Geneva: World Health Organization.
- Weathermon R, Crabb DW (1999) Alcohol and medication interactions. Alcohol Res Health 23(1): 40–54.